Author's response to reviews

Title: An unusual case of STEMI: bare metal stent fracture in native coronary artery.

Authors:

Giovanni GM Minardi (giovanni.minardi@libero.it)
Paolo G. PGP Pino (ppino@scamilloforlanini.rm.it)
Marco Stefano MSN Nazzaro (nazzaro@scamilloforlanini.rm.it)
Herribert HP Pavaci (herrpav@hotmail.com)
Martina MS Sordi (martinasordi@yahoo.it)
Cesare CG1 Greco (cesare.greco@uniroma1.it)
Carlo CG2 Gaudio (carlo.gaudio@tiscali.it)

Version: 2 Date: 20 January 2009

Author's response to reviews:

As requested, we report the information on patient's age, sex, and ethnic origin in the abstract.

Yours Sincerely,

Giovanni Minardi